Author's response to reviews

Title: A lifestyle intervention supported by mobile health technologies to improve the cardiometabolic risk profile of individuals at risk for cardiovascular disease and type 2 diabetes: Study rationale and protocol

Authors:

Melanie Stuckey (mstuckey@gmail.com)
Sheree Shapiro (sheree_shapiro@yahoo.ca)
Dawn Dill (dawn.gill@uwo.ca)
Robert J Petrella (petrella@uwo.ca)

Version: 2 Date: 29 October 2013

Author's response to reviews:

Dear Ms. Pafitis,

On behalf of the authors, “A lifestyle intervention supported by mobile health technologies to improve the cardiometabolic risk profile of individuals at risk for cardiovascular disease and type 2 diabetes: Study rationale and protocol”, I am pleased to acknowledge and attach evidence of funding and ethical approval for the protocol. We have also included the clinical trial registration number NCT01944124 at the end of the abstract as requested.

Sincerely,

Robert J. Petrella, MD, PhD